Author's response to reviews

Title: High-intensity mechanical therapy for loss of knee extension for worker's compensation and non-compensation patients

Authors:

Amanda L Dempsey (amandalaurendempsey@gmail.com)
Thomas P Branch (doctorbranch@yahoo.com)
Timothy Mills (tmills@university-ortho.com)
Robert M Karsch (rkarsch@university-ortho.com)

Version: 6 Date: 8 October 2010

Author's response to reviews:

We have written permission from the patient pictured in Figure 1 to use that image in this publication, and we will be happy to send an electronic copy of the patient’s written permission to the editorial staff if necessary. We have added the following text to the caption of Figure 1 as instructed:

Written informed consent was obtained from the patient for publication of this image. A copy of the written consent is available for review by the Editor-in-Chief of this journal.